UNIVERSITY^{OF} BIRMINGHAM

University of Birmingham Research at Birmingham

The BILAG2004-Pregnancy Index is a valid disease activity outcome measure for pregnant SLE patients

Yee, Chee-Seng; Khamashta, Munther; Akil, Mohammed; Kilding, Rachael; Giles, Ian; Gordon, Caroline

DOI:

10.1093/rap/rkac081

License

Creative Commons: Attribution (CC BY)

Document Version

Publisher's PDF, also known as Version of record

Citation for published version (Harvard):

Yee, C-S, Khamashta, M, Akil, M, Kilding, R, Giles, I & Gordon, C 2022, 'The BILAG2004-Pregnancy Index is a valid disease activity outcome measure for pregnant SLE patients', *Rheumatology Advances in Practice*, vol. 6, no. 3, rkac081. https://doi.org/10.1093/rap/rkac081

Link to publication on Research at Birmingham portal

General rights

Unless a licence is specified above, all rights (including copyright and moral rights) in this document are retained by the authors and/or the copyright holders. The express permission of the copyright holder must be obtained for any use of this material other than for purposes permitted by law.

•Users may freely distribute the URL that is used to identify this publication.

•Users may download and/or print one copy of the publication from the University of Birmingham research portal for the purpose of private study or non-commercial research.

•User may use extracts from the document in line with the concept of 'fair dealing' under the Copyright, Designs and Patents Act 1988 (?)

•Users may not further distribute the material nor use it for the purposes of commercial gain.

Where a licence is displayed above, please note the terms and conditions of the licence govern your use of this document.

When citing, please reference the published version.

Take down policy

While the University of Birmingham exercises care and attention in making items available there are rare occasions when an item has been uploaded in error or has been deemed to be commercially or otherwise sensitive.

If you believe that this is the case for this document, please contact UBIRA@lists.bham.ac.uk providing details and we will remove access to the work immediately and investigate.

Download date: 19. May. 2024

Rheumatology Advances in Practice, 2022, **00**, 1–4 https://doi.org/10.1093/rap/rkac081 Advance access publication 3 October 2022 **Original article**





Clinical science

The BILAG2004-Pregnancy Index is a valid disease activity outcome measure for pregnant SLE patients

Chee-Seng Yee (b) 1,*, Munther Khamashta², Mohammed Akil³, Rachael Kilding³, Ian Giles⁴, David Williams⁵, Ian N. Bruce (b) 6, Caroline Gordon (b) 7

Abstract

Objectives: This study was to determine whether the BILAG2004-Pregnancy Index (BILAG2004-P) has construct/criterion validity and is sensitive to change.

Methods: This was an observational multicentre study that recruited pregnant SLE patients. Data were collected on disease activity [using the BILAG2004-P and Physician Global Assessment (PGA)], investigations and therapy at each assessment. The overall BILAG2004-P score as determined by the highest score achieved by any system was used in the analysis. Cross-sectional analysis was used for construct and criterion validity. The comparison was with C3, C4 and anti-dsDNA for construct validity, while it was with change in therapy and PGA in criterion validity. Sensitivity to change was assessed by determining the relationship between the change in BILAG2004-P and the change in therapy between two consecutive visits.

Results: A total of 97 patients with 112 pregnancies were recruited. There were 610 assessments available for construct/criterion validity analysis (98.2% of pregnancies had more than one assessment) and 497 observations for sensitivity to change analysis. Increasing BILAG2004-P scores were associated with low C3. The active BILAG2004-P score (grade A or B) was associated with an increase in therapy and the PGA of active disease. There was an increasing likelihood of higher overall scores with an increase in therapy and the PGA of active disease. In the sensitivity to change analysis, an increase in the BILAG2004-P score was associated with an increase in therapy and inversely associated with an increase in therapy. A decrease in the BILAG2004-P score was associated with a decrease in therapy and inversely associated with an increase in therapy.

Conclusion: The BILAG2004-P has criterion validity and is sensitive to change.

Lay Summary

What does this mean for patients?

Lupus is a chronic autoimmune disease that mainly affects women of childbearing age. Thanks to improvements in lupus management, more women with lupus are able to become pregnant. However, changes during pregnancy can mimic some features of active lupus, which makes it more difficult to assess disease activity. Thus we have to adjust the way we assess lupus in pregnant patients. The BILAG2004-Pregnancy Index (BILAG2004-P) is a modified version of the BILAG-2004 index. While the BILAG-2004 Index is used to assess non-pregnant patients, the modified version should allow us to assess lupus activity in pregnant patients. This large study involved several medical centres and compared the BILAG2004-P with other markers and an external standard reference. We found that the BILAG2004-P appropriately measures lupus activity and is therefore a valid way to assess disease activity in pregnant patients. This finding will enable further research into lupus disease activity during pregnancy.

Keywords: BILAG2004-pregnancy, SLE, disease activity, pregnancy, construct validity, criterion validity, sensitivity to change

Introduction

As SLE affects mainly women of childbearing age, more women with SLE are becoming or potentially are able to become pregnant with improvement in the management of SLE. However, many pathophysiological features of pregnancy may mimic manifestations of SLE disease activity and

confound assessment of SLE disease activity during pregnancy. Therefore, standardized assessment of SLE disease activity outcome measures that are developed for the non-pregnant state need to be modified to take into account these changes before it can be used in pregnancy.

The BILAG2004-Pregnancy Index (BILAG2004-P) is a modification of the BILAG-2004 index for assessment of SLE

¹Department of Rheumatology, Doncaster and Bassetlaw Teaching Hospitals NHS Foundation Trust, Doncaster, UK

²Department of Women and Children's Health, King's College London, London, UK

³Department of Rheumatology, Sheffield Teaching Hospitals NHS Trust, Sheffield, UK

⁴Centre for Rheumatology Research, University College London, London, UK

⁵Department of Obstetrics, University College London Hospitals NHS Foundation Trust, London, UK

⁶Centre for Epidemiology Versus Arthritis, University of Manchester, Manchester, UK

⁷Rheumatology Research Group, University of Birmingham, Birmingham, UK

^{*}Correspondence to: Chee-Seng Yee, Department of Rheumatology, Doncaster Royal Infirmary, Armthorpe Road, Doncaster DN2 5LT, UK. E-mail: csyee@ymail.com

2 Chee-Seng Yee et al.

Key messages

- The BILAG2004-Pregnancy Index has criterion validity.
- The BILAG2004-Pregnancy Index is sensitive to change.
- The BILAG2004-Pregnancy Index is valid for assessment of SLE disease activity in pregnancy.

disease activity in pregnancy [1]. The modifications were to take into account pathophysiological changes in pregnancy. Most of the changes were to the glossary, with major changes in the scoring of the renal system. These modifications have been described in detail previously [1]. The development of this index enables standardized assessment of SLE disease activity outcomes during pregnancy. As it has a similar structure to the BILAG-2004 index [2], with nine system scores, it allows for a seamless transition in reporting between pregnant and non-pregnant states. A previous study showed the BILAG2004-P to be reliable [1]. The case record form, glossary and scoring scheme of this index are available as supplementary material (Supplementary Data S1-S3, available at Rheumatology Advances in Practice online). This analysis reports on the construct validity, criterion validity and sensitivity to change of the BILAG2004-P.

Methods

This was a multicentre study involving five centres in the UK that recruited SLE patients who were pregnant and fulfilled the revised ACR criteria for classification of SLE [3]. These five centres (Manchester, Birmingham, University College London, St Thomas' Hospital and Sheffield) had dedicated clinics looking after pregnant patients with rheumatic diseases. Patients were excluded from the study if they were <18 years of age or were unable to provide valid consent. This study was approved by the Trent Research Ethics Committee. It was carried out in accordance with the Helsinki Declaration and written informed consent was obtained from patients.

At each assessment, data were collected on disease activity [with BILAG2004-P and Physician Global Assessment (PGA)], investigations and therapy. The BILAG2004-P is a system-based index with nine ordinal system scores from grade A, indicating high disease activity, to grade E, indicating no disease activity and the system was not previously affected [1]. The overall BILAG2004-P score as determined by the highest score achieved by any individual system was used in the analysis. Grades D and E were combined for the analysis, as both indicate inactivity. A robust definition for change in treatment was used in the analysis that was similar to the one used in validation of the BILAG-2004 index (see Supplementary Data S4, available at *Rheumatology Advances in Practice* online) [2, 4].

All analyses were performed using Stata for Windows version 8 (StataCorp, College Station, TX, USA). Statistical analysis was with logistic regression and the robust estimator of variance was used to allow for correlation between assessments from the same patient. Results were reported as odds ratios (ORs) with 95% CIs.

Construct and criterion validity

This was a cross-sectional analysis. For construct validity, the overall BILAG2004-P score was compared with constructs that were C3, C4 and anti-dsDNA levels. The levels of C3, C4 and anti-dsDNA (by ELISA) were measured locally as part of routine practice using standard methodology by accredited laboratories. In criterion validity, the criterion for active disease used for comparison with the overall BILAG2004-P score were change in therapy and the PGA of active disease. The PGA of active disease was a yes/no response to the question 'Did this patient have active SLE disease?'. Change in therapy was classified into two categories: an increase in therapy and no increase in therapy. Although the scoring of the index was based on the physician's intention to treat, change in treatment had no bearing on the score of the index. Sensitivity, specificity, positive predictive value (PPV) and negative predictive value (NPV) were calculated for criterion validity analysis.

Sensitivity to change

In this longitudinal analysis, external responsiveness was used to assess the extent to which changes in the overall BILAG2004-P score over time relate to a corresponding change in therapy between two consecutive assessments [5]. Therefore, each observation for the analysis was derived from the changes that occurred between two consecutive assessments. There were three categories of change in therapy: no change, an increase in therapy and a decrease in therapy. Maximum-likelihood multinomial and binary logistic regression were used with change in therapy as the outcome variable and changes in the overall BILAG2004-P score as the explanatory variable. The baseline comparator for change in overall BILAG2004-P score was 'minimal change in score' as defined by 'no change in score' and change from grade D/E to C. The ability of change in overall BILAG2004-P score to predict a change in therapy was also assessed with the calculation of sensitivity, specificity, PPV and NPV.

Results

A total 97 pregnant SLE patients were recruited, who contributed 112 pregnancies. The ethnicity of the patients was 57.7% Caucasian, 18.6% South Asian, 17.5% African-Caribbean and 6.1% others. At the time of entry into the study for each pregnancy, the mean age was 31.3 years (s.d. 5.4) and the mean disease duration was 7.4 years (s.d. 5.1). This is summarized in Table 1.

Construct and criterion validity

There were 610 assessments available for this cross-sectional analysis (98.2% of pregnancies had more than one assessment). A total of 15.7% of assessments had active disease scores by the BILAG2004-P with overall scores of grade A or B and 21.5% of assessments were deemed to be active by the

Table 1. Baseline demographics of recruited patients (112 pregnancies from 97 patients)

Patient characteristics	Values
Ethnicity ($n = 97$ patients), %	
Caucasian	57.7
South Asian	18.6
African-Caribbean	17.5
Other	6.1
Age at recruitment ($n = 112$ pregnancies), years, mean (s.D.)	31.3 (5.4)
Disease duration at recruitment ($n = 112$ pregnancies), years, mean (s.d.)	7.4 (5.1)

Table 2. Distribution of disease activity across systems according to the BILAG2004-P (n = 610 assessments)

BILAG2004-P systems	BILAG2004-P grade, n (%)			
	A	В	С	D/E
Constitutional	0	0	9 (1.5)	601 (98.5)
Mucocutaneous	4(0.7)	54 (8.9)	96 (15.7)	456 (74.8)
Neuropsychiatric	1 (0.2)	1 (0.2)	0	608 (99.7)
Musculoskeletal	2 (0.3)	19 (3.1)	143 (23.4)	446 (73.1)
Cardiorespiratory	0	5 (0.8)	5 (0.8)	600 (98.4)
Gastrointestinal	0	0	0	610 (100)
Ophthalmic	0	0	0	610 (100)
Renal	3(0.5)	11 (1.8)	5 (0.8)	591 (96.9)
Haematological	1 (0.2)	4 (0.7)	204 (33.4)	401 (65.7)

Table 3. Logistic regression analysis comparing overall BILAG2004-P scores with an increase in therapy and the PGA of active disease

Overall BILAG2004-P score	Increase in therapy, OR (95% CI)	PGA active disease, OR (95% CI)
D/E (reference)	_	_
С	4.2 (1.3, 12.8)	37.7 (4.8, 292.6)
В	43.7 (14.1, 135.6)	1849.3 (192.1, 17800.8)
A	144.0 (35.5, 584.3)	∞

Table 4. Ability of active disease by overall BILAG2004-P score to predict an increase in therapy and the PGA of active disease

Calculation	Increase in therapy, % (95% CI)	PGA active disease, % (95% CI)	
Sensitivity	64.8 (51.8, 75.9)	66.4 (55.2, 76.0)	
Specificity	90.7 (86.5, 93.7)	98.1 (94.8, 99.3)	
PPV	47.9 (37.6, 58.4)	90.6 (76.5, 96.6)	
NPV	95.1 (92.8, 96.7)	91.4 (87.2, 94.3)	

BILAG2004-P: BILAG2004-Pregnancy Index; PGA: Physician's Global Assessment.

PGA. The distribution of disease activity across the systems is summarized in Table 2. The prevalence of low C3, low C4 and elevated anti-dsDNA were 12.2% (of 510 assessments), 21.8% (of 515 assessments) and 22.0% (of 505 assessments), respectively. There was an increase in therapy in 11.6% of assessments while a decrease in therapy occurred in 7% of assessments.

Analysis with ordinal logistic regression showed that there was a significant association between increasing overall

Table 5. Cross-tabulation of changes in the overall BILAG2004-P score against the change in therapy longitudinally (*n* = 497 observations)

Change in overall BILAG2004-P score	Change in therapy, n (%)		
	Decrease	No change	Increase
Minimal change	36 (7.2)	285 (57.3)	33 (6.6)
Decrease to C/D	20 (4.0)	79 (15.9)	1 (0.2)
Decrease to B	2 (0.4)	2 (0.4)	1 (0.2)
Increase to B	0	14 (2.8)	18 (3.6)
B to A	0	1 (0.2)	1 (0.2)
C/D to A	0	0	4 (0.8)

Table 6. Ability of increase in the overall BILAG2004-P score to predict an increase in therapy

Calculation	% (95% CI)	
Sensitivity	39.7 (4.4)	
Specificity	96.6 (94.6, 97.9)	
PPV	40.0 (30.0, 51.0)	
NPV	93.8 (90.6, 95.9)	

BILAG2004-P score with low C3 [OR 2.0 (95% CI 1.1, 3.5)] but not with low C4 [OR 1.0 (95% CI 0.6, 1.9)] or elevated anti-dsDNA [OR 1.2 (95% CI 0.7, 2.1)].

The active overall BILAG2004-P score (grade A or B) was significantly associated with an increase in therapy [OR 18.0 (95% CI 10.3, 31.5)] and the PGA of active disease [OR 103.0 (95% CI 33.0, 322.2)]. There was an increasing likelihood of a higher overall score with an increase in therapy and the PGA of active disease (Table 3). There was total agreement between a grade A score and the PGA of active disease. Table 4 summarizes the ability of the active overall BILAG2004-P score in predicting an increase in therapy and the PGA of active disease. There was good specificity and NPV.

Sensitivity to change

There were 497 observations available for this longitudinal analysis. An increase in therapy was observed in 11.7% of observations and a decrease in therapy occurred in 11.7% of observations. The distribution of changes in the overall BILAG2004-P score is summarized in Table 5.

Multinomial logistic regression analysis showed that an increase in the overall BILAG2004-P score was significantly associated with an increase in therapy [OR 13.2 (95% CI 5.8, 30.1)] and was inversely associated with a decrease in therapy [OR <0.1 (95% CI <0.1, <0.1)]. On the other hand, a decrease in the overall BILAG2004-P score was significantly associated with a decrease in therapy [OR 2.2 (95% CI 1.3, 3.7)] and was inversely associated with an increase in therapy [OR 0.2 (95% CI <0.1, 0.96)]. An increase in therapy was unlikely in the absence of an increase in the overall BILAG2004-P score, as evident by the excellent specificity and NPV (Table 6).

Discussion

This was a relatively large multicentre study assessing the construct validity, criterion validity and sensitivity to change of the BILAG2004-P. However, most of the assessments had

4 Chee-Seng Yee et al.

low-level disease activity (grade C, D or E scores), with >75% of assessments (for construct and criterion validity analysis) and observations (for sensitivity to change analysis) having no change in therapy. Despite this, we demonstrated that the BILAG2004-P has criterion validity from its significant association with both an increase in therapy and the PGA of active disease. The longitudinal analysis showed that it is sensitive to change through the relationship of change in the index score with change in therapy. In the analysis of construct validity, the only significant association demonstrated was with low C3.

The major limitation of this study is the low prevalence of active disease as noted in the small numbers of grade A and B scores. Only 2.3% of the assessments had active renal disease (grade A or B), which was shown to have the strongest association with low C3/C4 and elevated anti-dsDNA [6–8]. It is to be expected that most pregnant SLE patients would have minimal disease activity, as current management of SLE patients is to minimize the risk of active disease before and during pregnancy in order to improve pregnancy outcome [9]. This may explain the inability to demonstrate the association between active disease scores and low C4 or elevated anti-dsDNA.

In addition, most of the disease activity affected the mucocutaneous, musculoskeletal and renal systems, thus it is not possible to assess the index through its full range of disease activity with this dataset. Despite this, we do not anticipate any issues with regards to the validity of this index, as this index was modified (specifically for use in pregnancy) from the BILAG-2004 index, which was comprehensively validated [2, 4, 10]. As such, it would be reasonable to conclude that the BILAG2004-P is valid for use to assess SLE disease activity during pregnancy.

In conclusion, the BILAG2004-P will be the only SLE disease activity index that has undergone a thorough validation process in pregnant SLE patients, as it has been shown to be reliable previously [1], and this study has demonstrated construct/criterion validity along with sensitivity to change. There are other disease activity indices that have been used to assess SLE disease activity during pregnancy, such as the SLE-DAS [11] and SLEPDAI [12], but they have not been validated for use in pregnancy. Unlike the BILAG2004-P, which is a system-based disease activity index, these other indices are global disease activity indices that summarize disease activity into a single global score. The majority of these other indices are based on modifications of existing indices that were developed for non-pregnant states to take into account pathophysiological changes in pregnancy. These other indices have been discussed succinctly in a review article [12].

Supplementary data

Supplementary data are available at *Rheumatology Advances* in *Practice* online.

Data availability statement

The data underlying this article will be shared upon reasonable request to the corresponding author.

Funding

This study was supported by a grant from Versus Arthritis (16081).

Disclosure statement: The authors have declared no conflicts of interest.

Acknowledgements

We would like to thank Mark Kilby, Tracey Johnston, Fiona Fairlie and Maggie Blott.

References

- Yee CS, Akil M, Khamashta M et al. The BILAG2004-Pregnancy index is reliable for assessment of disease activity in pregnant SLE patients. Rheumatology 2012;51:1877–80.
- Yee CS, Farewell V, Isenberg DA et al. The BILAG-2004 index is sensitive to change for assessment of SLE disease activity. Rheumatology 2009;48:691–5.
- Hochberg MC. Updating the American College of Rheumatology revised criteria for the classification of systemic lupus erythematosus. Arthritis Rheum 1997;40:1725.
- Yee CS, Farewell V, Isenberg DA et al. British Isles Lupus Assessment Group 2004 index is valid for assessment of disease activity in systemic lupus erythematosus. Arthritis Rheum 2007;56: 4113–9
- Husted JA, Cook RJ, Farewell VT, Gladman DD. Methods for assessing responsiveness: a critical review and recommendations. J Clin Epidemiol 2000;53:459–68.
- Swaak AJG, Groenwold J, Aarden LA, Statius van Eps LW, Feltkamp EW. Prognostic value of anti-dsDNA in SLE. Ann Rheum Dis 1982;41:388–95.
- Swaak AJG, Groenwold J, Bronsveld W. Predictive value of complement profiles and anti-dsDNA in systemic lupus erythematosus. Ann Rheum Dis 1986;45:359–66.
- ter Borg EJ, Horst G, Hummel EJ, Limburg PC, Kallenberg CGM. Measurement of increases in anti-double-stranded DNA antibody levels as a predictor of disease exacerbation in systemic lupus erythematosus. Arthritis Rheum 1990;33:634–43.
- Flint J, Panchal S, Hurrell A et al. BSR and BHPR guideline on prescribing drugs in pregnancy and breastfeeding—part I: standard and biologic disease modifying anti-rheumatic drugs and corticosteroids. Rheumatology 2016;55:1693–7.
- Yee CS, Farewell V, Isenberg DA et al. Revised British Isles Lupus Assessment Group 2004 index: a reliable tool for assessment of systemic lupus erythematosus activity. Arthritis Rheum 2006;54: 3300–5.
- 11. Larosa M, Costedoat-Chalumeau N, Guettrot-Imbert G *et al.* SLE-DAS in the first trimester of gestation predicts maternal lupus flares later in pregnancy. Front Pharmacol 2021;12:660123.
- Buyon JP, Kalunian KC, Ramsey-Goldman R et al. Assessing disease activity in SLE patients during pregnancy. Lupus 1999;8: 677–84





A 2nd generation, JAK1 preferential inhibitor for moderate to severe RA1-6

While 1st generation JAK inhibitors are relatively non-selective,²⁻⁶ JYSELECA has over 5x greater potency for JAK1 over JAK2/3 and TYK21*

Balancing sustained efficacy⁷⁻¹¹ with acceptable tolerability^{1,12}



Indicated for the treatment of moderate to severe active rheumatoid arthritis in adult patients who have responded inadequately to, or who are intolerant to one or more disease modifying anti-rheumatic drugs. May be used as monotherapy or in combination with methotrexate.1

*From biochemical assays, the clinical relevance of which is uncertain. JAK, Janus kinase; RA, rheumatoid arthritis; TYK, tyrosine kinase.

Refer to Summary of Product Characteristics (SmPC) before prescribing, and for full prescribing information.

Refer to Summary of Product Characteristics (SmPC) before prescribing, and for full prescribing information.

INSELECA® Tigotinib 100 mg or 200 mg film-coated tablets.

Indication: yyseleca is indicated for the treatment of moderate to severe active rheumatoid arthritis in adult patients who have responded inadequately to, or who are intolerant to one or more disease modifying anti rheumatic drugs (DMARDs). yseleca may be used as monotherapy or in combination with methotrexate (MTX). Dosage: Adults; 200 mg once daily. Taken orally with/without food. Its recommended that tablets are swallowed whole. Laboratory Monitoring: Refer to the SmPC for information regarding laboratory monitoring and dose initiation or interruption. Elderly: A starting dose of 100 mg once daily is recommended for patients aged 75 years and older as clinical experience is limited. Renal impairment. No dose adjustment required in patients with estimated creatinine clearance (CrCl) ≥ 60 mL/min. A dose of 100 mg of filgotinib once daily is recommended for patients with moderate or severe renal impairment (CrCl 15 to < 60 mL/min). Not recommended in patients with CrCl < 15 mL/min. Hepatic impairment: Mild/moderate hepatic impairment: not dose adjustment required. Severe hepatic impairment: no dose adjustment required. Severe hepatic impairment: not recommended. Children (< 18years): Safety and efficacy not yet established. Contraindications: Hypersensitivity to the active substance or to any of the excipients. Active tuberculosis (TB) or active serious infections. Pregnancy. Warnings/Precautions: See SmPC for full information. Immunosuppression combination use, with immunosuppression cannot be excluded. Infections: Infections, including serious infections such as pneumonia and opportunistic infections eg. unberculosis (TB), oesophageal candidiasis, and cryptococcosis have been reported. Risk benefit should be assessed prior to initiating in patients with risk factors for infections (see SmPC). Patients should be closely monitored for the devel lave been reported. Kisk beneart should be assessed phot to nitiating in patients with risk factors for infections (see SmPC). Patients should be closely monitored for the development of igns and symptoms of infections during and after figlotnib reatment. Treatment should be interrupted if the patient

is not responding to antimicrobial therapy, until infection is controlled. There is a higher incidence of serious infections in the elderly aged 75 years and older, caution should be used when treating this population. <u>Tuberculosis</u>: Patients should be screened for TB before initiating filgotinib, and filgotinib should not be administered to patients with active TB. <u>Virial reactivation</u>: Cases of herpes virus reactivation (e.g., herpes zoster), were reported in clinical studies (see SmPC). If a patient develops herpes zoster, filgotinib treatment should be temporarily interrupted until the episode resolves. Screening for viral hepatitis and monitoring for reactivation should be performed. <u>Malignancy</u>: Immunomodulatory medicinal products may increase the risk of malignancies. Malignancies were observed in clinical studies (see SmPC). <u>Fertility</u>: In animal studies, decreased fertility, impaired spermatogenesis, and histopathological effects on male reproductive organs were observed (see SmPC). The potential effect of filgotinib on sperm production and male fertility in humans is currently unknown. <u>Haematological abnormalities</u>: Do not start therapy, or temporarily stop, if Absolute Neutrophil Count (ANC) 1 × 10° (ells]/L, LIC CoJS × 10° cells/L or haemoglobin <8 g/dL. Temporarily stop therapy if these values are observed during routine patient management. <u>Vaccinations</u>: Use of live vaccines during, or immediately prior to, filgotinib treatment is not recommended. <u>Lipids</u>: Treatment with filgotinib was associated with dose dependent increases in lipid parameters, including total cholesterol, and high-density lipoprotein (LDL) levels, while low density lipoprotein (LDL) levels were slightly increased (see SmPC). <u>Cardiovascular risk</u>: Rheumatoid arthritis patients have an increased risk for cardiovascular disorders. Patients should have risk factors (e.g., hypertension, hyperlipidaemia) managed as part of usual standard of care. <u>Venous thrombosis</u> (DVT) and pulmonary embolism (PE) have been reported in pati of DVT/PE, or patients undergoing surgery, and prolonged

Learn more at strengthofbalance.co.uk

immobilisation. <u>Lactose content:</u> Contains lactose; patients with rare hereditary problems of galactose intolerance, total lactase deficiency or glucose-galactose malabsorption should not take filgotinib. **Pregnancy/Lactation:** Filgotinib is contraindicated in pregnancy. Filgotinib should not be used during breast-feeding. Women of childbearing potential must use effective contraception during and for at least 1 week after cessation of treatment. **Driving/Using machinery:** No or negligible influence, however dizziness has been reported. **Side effects:** See SmPC for full information. <u>Common (21/100 to <1/100)</u>: herpes coster, pneumonia, neutropenia, hypercholesterolaemia infection and dizziness. <u>Uncommon (21/1000 to <1/100)</u>; herpes zoster, pneumonia, neutropenia, hypercholesterolaemia and blood creatine phosphokinase increase. Serious side effects: See SmPC for full information Legal category: POM Pack: 30 film-coated tablets/bottle Price: UK Basic NHS cost: £863.10 Marketing authorisation number(s): <u>Great Britain</u> Jyseleca 100mg film-coated tablets PLGB 42147/0002 Northern Ireland Jyseleca 100mg film-coated tablets EU/1/20/1480/003 EU/1/20/1480/003 EU/1/20/1480/004 Further information: Galapagos UK, Belmont House, 148 Belmont Road, Uxbridge UB8 10S, United Kingdom 00800 7878 1345 medicalinfo@glpg.com Jyseleca® is a trademark. **Date of Preparation**: January 2022 UK-RA-FIL-202201-00019

Additional monitoring required

Adverse events should be reported.

For Great Britain and Northern Ireland, reporting form and information can be found at <u>yellowcard.mhra.gov.</u> and information can be found at <u>yeutowcara.mnra.gov.u</u> or via the Yellow Card app (download from the Apple Ap Store or Google Play Store). Adverse events should also be reported to Galapagos via email to DrugSafety.UK.Ireland@glpg.com or 00800 7878 1345

References: 1. JYSELECA SPC. Available at: www.medicines.org.uk. Last accessed: June 2022. 2. Angelini J, et al. Biomolecules 2020;10(7):E1002. 3. Banerjee S, et al. Drugs 2017;77:521–546. 4. O'Shea JJ, et al. Nat Rev Rheumatol 2013;9(3):173–182. 5. Traves PG, et al. Ann Rheum Dis 2021;0:1-11. 6. McInnes IB, et al. Arthr Res Ther 2019;21:183. 7. Combe B, et al. Ann Rheum Dis 2021;doi:10.1136/annrheumdis-2020-219214. 8. Genovese MC, et al. JAMA 2019;322 (4):315–325. 9. Westhovens R, et al. Ann Rheum Dis 2021;doi:10.1136/annrheumdis-2020-219213. 10. Combe B, et al. Arthritis Rheumatol 2021;73(supp) 10). https://acrabstract/clinical-outcomes-up-to-week-48-of-ongoing-long-term-extension-trial-of-a-patients-with-inadequate-response-to-mtx-initially-treated-with-filgotinib-ra-dalimumab-during-th/. Last accessed: June 2022. 11. Buch MH, et al. Arthritis Rheumatol 2021;73 (suppl 10). https://acrabstracts.org/abstract/clinical-outcomes-up-to-week-48-of-ongoing-filgotinib-ra-plase-3-trial/. Last accessed: June 2022. 12. Winthrop K, et al. Arthritis Rheumatol 2021;73(suppl 10). Available at: https://acrabstracts.org/abstract/integrated-safety-analysis-update-for-filgotinib-in-patients-with-moderately-to-severely-active-rheumatoid-arthritis-receiving-treatment-over-a-median-of-2-2-years/. Last accessed: June 2022.

Galápagos

June 2022 GB-RA-JY-202205-00033

JYSELECA, GALAPAGOS and the JYSELECA and GALAPAGOS logos are registered trademarks of Galapagos NV. © 2022 Galapagos NV. All rights reserved.

